Author’s response to reviews

Title: Coverage of Jade Goody’s cervical cancer in UK newspapers: a missed opportunity for health promotion?

Authors:

Shona Hilton (shona@msoc.mrc.gla.ac.uk)
Kate Hunt (kate@msoc.mrc.gla.ac.uk)

Version: 4 Date: 15 June 2010

Author’s response to reviews:

Dear BMC Series Editorial Production Team,
I have attached and made changes to our paper (v14) in light of your recommendations.
We have now ensured that the title is identical in the manuscript and the submission system.
We have included Figure 1 in the manuscript and are grateful for you noticing that it was missing (now on page 23)
Minor revisions we have made include:
Authors’ contributions: has been removed these from the title page.
Keywords and running title has also been removed.
The table and figure is now after the reference list.
However, I have not managed to remove the visible vertical lines from the tables,
Kind regards,
Shona Hilton and Kate Hunt